Author’s response to reviews

Title: Infringement of the right to surgical informed consent: Negligent disclosure and its impact on patient trust in surgeons at public general hospitals - The voice of the patient

Authors:

Gillie Gabay (gillie.gabay@gmail.com)

Yaarit Bokek-Cohen (ybokek@gmail.com)

Version: 5 Date: 04 Sep 2019

Author’s response to reviews:

Hopefully. I understood the requests.

The statement "Written informed consent was obtained from all participants", was moved to the ethics approval sections.

I removed bold text, typically stressing the tone of participants in qualitative studies - hope that is what you meant. Please advise should you have a different intention.

Under sharing raw data, I explained that I do not have permission to share raw data.

Thank you.